Author's response to reviews

Title: Reverse LISS plating for intertrochanteric hip fractures in elderly patients

Authors:

CQ Zhang (zhangchqin@yahoo.cn)
Y Sun (sycedar@hotmail.com)
DX Jin (david1979982@126.com)
C Yao (goldeyeboy@gmail.com)
SB Chen (chenshengbao@yahoo.com.cn)
BF Zeng (bingfangz@yahoo.com.cn)

Version: 7 Date: 20 July 2010

Author’s response to reviews:

Dear editor,

Thank you for the comments. According to the requirements, we have made the revision. Please check them, should you have any questions, contact us without hesitation.